Author's response to reviews

Title: Evaluation of CXCL9 and CXCL10 as circulating biomarkers of human cardiac allograft rejection

Authors:

Kristjan Karason (kristjan.karason@medfak.gu.se)
Margareta Jernas (Margareta.jernas@medic.gu.se)
Daniel A Hagg (Daniel.hagg@medic.gu.se)
Per-Arne Svensson (Per-Arne.Svensson@medic.gu.se)

Version: 4 Date: 15 June 2006

Author’s response to reviews:

1) The website addresses have been moved to the reference list and removed from the manuscript text. However, there were problems with the reference list. The web addresses would not be extracted after upload. I have discussed this with Robin Stewart and he said that I for the time being should ignore that the references were not correctly extracted.

2) The figure and tables legends have been moved to after the references.